

Citation: Hernandez M, Ramon-Julvez U, Ferraz F, with the ADNI Consortium (2022) Explainable AI toward understanding the performance of the top three TADPOLE Challenge methods in the forecast of Alzheimer's disease diagnosis. PLoS ONE 17(5): e0264695. https://doi.org/10.1371/journal. pone.0264695

Editor: Nguyen Quoc Khanh Le, Taipei Medical University, TAIWAN

Received: August 20, 2021

Accepted: February 16, 2022

Published: May 6, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0264695

Copyright: © 2022 Hernandez et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The authors collected data from the ADNI repository (<u>http://adni.loni.usc.</u>edu). Researchers are able to access to these data

RESEARCH ARTICLE

Explainable AI toward understanding the performance of the top three TADPOLE Challenge methods in the forecast of Alzheimer's disease diagnosis

Monica Hernandez[®]*, Ubaldo Ramon-Julvez, Francisco Ferraz, with the ADNI Consortium¹¹

Aragon Institute on Engineering Research, University of Zaragoza, Zaragoza, Spain

¶ Membership of the ADNI Consortium is listed in the Acknowledgments. * mhg@unizar.es

Abstract

The Alzheimer's Disease Prediction Of Longitudinal Evolution (TADPOLE) Challenge is the most comprehensive challenge to date with regard to the number of subjects, considered features, and challenge participants. The initial objective of TADPOLE was the identification of the most predictive data, features, and methods for the progression of subjects at risk of developing Alzheimer's. The challenge was successful in recognizing tree-based ensemble methods such as gradient boosting and random forest as the best methods for the prognosis of the clinical status in Alzheimer's disease (AD). However, the challenge outcome was limited to which combination of data processing and methods exhibits the best accuracy; hence, it is difficult to determine the contribution of the methods to the accuracy. The quantification of feature importance was globally approached by all the challenge participant methods. In addition, TADPOLE provided general answers that focused on improving performance while ignoring important issues such as interpretability. The purpose of this study is to intensively explore the models of the top three TADPOLE Challenge methods in a common framework for fair comparison. In addition, for these models, the most meaningful features for the prognosis of the clinical status of AD are studied and the contribution of each feature to the accuracy of the methods is quantified. We provide plausible explanations as to why the methods achieve such accuracy, and we investigate whether the methods use information coherent with clinical knowledge. Finally, we approach these issues through the analysis of SHapley Additive exPlanations (SHAP) values, a technique that has recently attracted increasing attention in the field of explainable artificial intelligence (XAI).

1 Introduction

Alzheimer's disease (AD) involves a progressive deterioration of neuronal structures and brain function, leading to severe cognitive impairment, dementia, and, ultimately, death [1]. The most consistent risk factor for developing AD is advancing age [2, 3]. Owing to the estimated

in the same way as the authors did. The authors do not have special access privileges that others would not have.

Funding: 1) MH, URJ Grant number: PID2019-104358RB-I00 Funder: Ministerio de Ciencia e Innovacion URL: <u>https://www.ciencia.gob.es/site-</u> web/;jsessionid=

FE84CCA1BA4EAA27EABCADF4007CCF07 The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. 2) MHG Grant number: T 64 20R Funder: Gobierno de Aragon URL: https://www. aragon.es/ The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AD, Alzheimer's Disease; ADAS, Alzheimer's Disease Assessment Scale: ADNI. Alzheimer's Disease Neuroimaging Initiative; APOE4, Apolipoprotein-E4; BAIPETNMRC, Fluorodeoxyglucose Positron Emission Tomography measures; BCA, Balanced Classification Accuracy; bl, baseline; CDR, Clinical Dementia Rating; CN, Cognitive Normal; CSF, Cerebrospinal Fluid; DTI, Diffusion Tensor Imaging; DX, Diagnosis; ECog, Everyday Cognition; FAQ, Functional Activities Questionnaire; FDG, Fluorodeoxyglucose; HCI, Hypometabolic Convergence Index; M, months from baseline; mAUC, multi-class Area Under the receiver operating Curve; MCI, Mild Cognitive Impairment; MMSE, Mini-Mental State Evaluation; MOCA, MOntreal Cognitive Assessment; MRI, Magnetic Resonance Imaging; PET, Positron Emission Tomography; RAVLT, Rey Verbal Learning Test; RF, Random Forest; SHAP, SHapley Additive exPlanations; ST*CV, cortical parcellation volume of FreeSurfer segmentation; ST*SA, surface area of FreeSurfer segmentation; ST*SV, white matter parcellation volume of FreeSurfer segmentation; ST*TA, cortical thickness average of FreeSurfer segmentation; ST*TS, cortical thickness standard deviation of FreeSurfer segmentation; SVM, Support Vector Machines; TADPOLE, Alzheimer's Disease Prediction Of Longitudinal Evolution; UCBERKELEYAV45, ADNI summary table with AV45 standardized uptake value ratios for 18Fflutemetamol PET; UCSFFSL, ADNI summary table with MRI longitudinal biomarkers; UCSFFSX, ADNI summary table with MRI cross-sectional biomarkers; UPENNBIOMK9, ADNI summary with CSF puncture results; XAI, Explainable Artificial Intelligence; XGB, Extreme Gradient Boosting.

growth rate of the global population aged 65 and above in future decades, a considerable increase is expected in the proportion of people suffering from AD.

Clinical research on treatments against AD is approached in two different directions [4, 5]. Some researchers are working toward developing treatments that slow down or reverse the loss of cognitive abilities. Other researchers are working toward developing ways to diagnose the cause of dementia as early as possible in order to design preventive therapies. To date, the development of effective protective or preventive therapies has been hindered by limited knowledge on the causes and mechanisms of the neurodegenerative processes underlying AD. The community agrees on the importance of early diagnosis and accurate prognosis in the success of either the protective or the preventive approach to the treatment of AD [6].

To conduct diagnosis, clinical experts must interpret high-dimensional multi-modal data, including the clinical history of the patients, the outcomes of different cognitive tests, brain scans of different imaging modalities, and mutations of genetic information [7]. Diagnosis is known to be highly subjective. Beach et al. reported a sensitivity range of 70.9 to 87.3 and a specificity range of 44.3 to 70.8 in the identification of probable AD individuals [8]. Postmortem analyses have reported errors in diagnosis of up to 20% of cases [9]. The prognosis of progression to a more severe condition is even a harder problem.

Computational systems for facilitating the diagnosis and prediction of patient condition may enable clinicians to identify the best treatment options in a personalized manner and assess the changes in disease indicators owing to the effect of treatments [10]. Artificial intelligence (AI) is playing an increasingly relevant role with the emergence of computer-aided systems and compelling tools for performing diagnosis and predicting disease evolution [11, 12]. In recent decades, several initiatives have been conducted in the form of challenges to identify the best-performing systems for the diagnosis or prognosis of AD from different types of biomarkers [13–15].

Among them, The Alzheimer's Disease Prediction Of Longitudinal Evolution (TADPOLE) Challenge is probably the most comprehensive challenge to date with regard to the number of subjects, considered features, and challenge participants [15, 16]. TADPOLE deals with the problem of prognosis. The challenge focuses on forecasting the trajectories of three key features, namely clinical status, cognitive decline, and ventricle atrophy, over a five-year period. The initial objective of TADPOLE was the identification of the best-performing data processing, features, and methods for the prediction of the progression of subjects at risk of developing AD. The challenge provided a dataset from the Alzheimer's Disease Neuroimaging Initiative (ADNI) with a complete history of measurements obtained from demographics, cognitive tests, different imaging modalities, and genetics. In addition, it provided a dataset with forecast measurements in real time, which constitutes an actual evaluation set for AD evolution. In the following, we focus on the problem of clinical status prediction.

The results of the TADPOLE Challenge were revealing and identified tree-based ensemble methods such as gradient boosting and random forest in the first positions for the prognosis of the clinical status [17]. Unexpectedly, the highest position for a neural network method was achieved by a long-short term memory network (LSTM), i.e., the ninth position. Feature importance was studied globally, using the outcomes of all the challenge methods. Amyloid beta measured from the cerebrospinal fluid (CSF) puncture, image features from DTI, and APOE status contributed toward the highest accuracy. Meanwhile, image features from tau, amyloid PET, and FDG-PET contributed toward the worst accuracy. The data processing showed considerable variability with differences in record selection, data encoding, augmentation, and imputation of missing values proposed by the challenge participants.

Owing to the challenge design, TADPOLE provided answers limited to identifying which combination of data processing, feature selection, and methods exhibits the best accuracy

metrics. The variability in data preprocessing and feature selection may contribute toward better or worse accuracy of the methods; hence, it is difficult to determine the contribution of the methods to the accuracy. The same problem arises in the assessment of the importance of a given feature in disease diagnosis. In addition, the challenge focused mainly on improving performance while ignoring interpretability issues one more time. It is well acknowledged that the patterns learned using complex and accurate models may not be coherent with clinical knowledge. Therefore, accurate methods should be provided with a set of easily understandable explanations as to why or how the method reaches a specific decision and whether this reasoning is coherent with the knowledge applied in clinical practice.

To obtain deeper insights into the performance of the TADPOLE Challenge methods, it is necessary to perform a comparative assessment of the different models generated by the methods under the same data preprocessing choices. In addition, it would be of interest to study the importance given by each method to the features as well as to establish whether they are consistent with clinical knowledge in order to gain a better understanding of the mechanisms involved in the correct or incorrect diagnosis and predictions. Thus, the purpose of this study is to delve into the following questions using explainable artificial intelligence (XAI) [18–20]:

- Why did the best methods achieve the best accuracy metrics?
- How can we quantify the contribution of each feature toward achieving the best accuracies?
- Which features were the most meaningful for the best methods?
- Do the best methods use information coherent with clinical knowledge?

The answers to these questions are relevant to increasing or decreasing trust in the best-performing TADPOLE Challenge methods for the prognosis of AD toward realizing actionable systems useful in clinical practice.

Our study focuses on the top three methods, namely gradient boosting (*Frog*), random forest (*ThreeDays*), and support vector machine (*EMC-EB*). We built three models trained with balanced versions of the original TADPOLE training set. We performed the same preprocessing of the data for the three models for fair comparison of the methods. As a first approach toward model interpretability in this problem, we studied the importance given by the models to the features by analyzing the system built-in importance, when available, and the SHapley Additive exPlanations (SHAP) values, a recent technique proposed for XAI [20–22]. Both built-in importance and SHAP have been shown to be suitable for quantifying the contribution of a given feature toward the obtained accuracy. In addition, the various ways of representing the information provided by SHAP explainers enabled us to establish whether the models can use feature values that are coherent with clinical knowledge.

Explainable AI has recently started to provide interesting insights into machine learning methods for the diagnosis of AD. Within the TADPOLE Challenge framework, Moore et al. [17] used permutation-based importance in random forests. They found that the diagnosis, participant identifier, and time delay were the most important features. On the basis of these results, the authors argued that the method worked as expected, as the best predictor of future diagnosis should be the last diagnosis, especially for short time horizons, and time should help modulate the confidence in the prevalence of the last diagnosis label. Nguyen et al. [23] used feature ablation to establish the impact of different features on the prediction performance. The authors found that ablating diagnosis resulted in the most significant drop in the evaluation metrics. The drop obtained by ablating the Clinical Dementia Rating-–Sum of Boxes measurement (CDR-SB) was also significant.

Outside the TADPOLE Challenge framework, El-Sappagh et al. [24] recently proposed the use of SHAP values in a random-forest-based multilayer multimodal detection and prediction model. The detection layer performs classification into Alzheimer's disease, mild cognitive impairment, and cognitive normal subjects (AD/MCI/CN). The prediction layer establishes the distinction between stable and progressive MCI subjects (sMCI/pMCI) in a three-year window. Our study differs from El-Sappagh's work in the following aspects. First, our study is conducted within the TADPOLE Challenge framework; therefore, the data used for training, testing, and evaluation are those established by the challenge organizers. El-Sappagh et al. reduced the feature set to a total of 28 baseline features through automatic feature selection and manual expert verification; features outside the TADPOLE Challenge were included among the 28 features. The interpretability study was reduced to random forests. By contrast, we consider a total of 1818 features from both baseline and follow-up visits. We focus on the problem of prognosis, as it is the objective of the TADPOLE Challenge. We present an interpretability study for three methods, including random forests. Both El-Sappagh's study and our study make valuable contributions toward providing a complete understanding of the problem through interesting methods recently proposed in the emerging field of XAI. These systems may be combined with more sophisticated methods for dealing with multi-modal information toward realizing truly actionable explainable systems [25].

The remainder of this paper is organized as follows. Section 2 describes the datasets and data processing methods used in our study. Sections 3 and 4 describe the methods and evaluation metrics considered in our work. Section 5 presents the results of our study. Section 6 provides an analysis of our results in the context of the findings in the state of the art. Finally, Section 7 discusses the most remarkable results of our work and draws some interesting conclusions.

2 TADPOLE Challenge dataset: Description and processing

The training and test sets used in this study were built from the longitudinal dataset provided by the TADPOLE Challenge organizers, namely D1 and D2 data (<u>https://tadpole.grand-challenge.org/Data</u>). In addition, the future set D4 was used for evaluation. From D2, the last record from each patient in D4 was reserved as the test data and all the records from the test subjects were removed from the training set candidates to avoid data leakage.

The original sets include 12 734 samples for training, and 219 and 234 samples for testing and evaluation, respectively. We removed from the training set any record from the subjects in the test set. Further, we balanced the training set by a random under-sampling of the over-represented diagnosis groups. The resulting sets include 7 001 samples for training, and 219 and 210 samples for testing and evaluation, respectively. Table 1 lists the number of subjects for the different diagnosis groups in the data-leakage-corrected and balanced training, test, and evaluation datasets.

Table 1. Number of cognitive normal (CN), mild cognitive impairment (MCI), and Alzheimer's disease (AD) individuals in the training, test, and evaluation sets used in this work.

	train (corrected, D1-D2)	train (balanced, D1-D2)	test (D2)	eval (D4)
CN	2730	2337	106	86
MCI	4141	2337	98	92
AD	2337	2337	15	32
Total	9208	7001	219	210

https://doi.org/10.1371/journal.pone.0264695.t001

The clinical criteria for diagnosis in ADNI can be found in [26]. First, memory complaints were recorded on the basis of cognitive exams. The CN subjects had no complaints, whereas the MCI and AD subjects had complaints. Then, the mini-mental state examination (MMSE) results were observed. These results ranged from 24 to 30 for CN and MCI subjects and from 20 to 26 for AD subjects. The clinical dementia rating (CDR) score was 0 for CN subjects, 0.5 for MCI subjects, and 0.5 or 1 with a mandatory requirement of the memory box score for AD subjects. Finally, the Wechsler Memory Scale-Revised (WMS-R) was used with cutoff scores based on education. Subjects with scores coherent with a given group were selected to participate in the ADNI cohort. From these criteria, the TADPOLE Challenge includes CDRSB_bl, CDRSB, MMSE_bl, and MMSE as features where bl denotes baseline measurements.

The training and test sets were processed using TADPOLE-SHARE code (https://tadpoleshare.github.io). The diagnosis column was created from the feature DXCHANGE according to TADPOLE-SHARE convention. As a result, the records were categorized into CN, MCI, and AD labels. After this labeling, a considerable number of records (hundreds) have an empty diagnosis. We performed imputation of these labels using nearest-neighbor interpolation based on the patient closest early time points. Seven records with a single visit and no clinical diagnosis were excluded from the original training set, and 24 patients without clinical diagnosis were excluded from the original evaluation set. Our imputation of diagnosis differs from that used in TADPOLE-SHARE, thus obtaining a larger amount of valuable training data. Then, categorical information was replaced by numerical values, and a total of 89 features evidently irrelevant to the diagnosis problem (e.g., update stamp dates) were manually selected and removed, yielding training and test sets with 1818 features. Finally, empty values were imputed using the values of the closest early time points or the mean of the training set data when nearest-neighbor interpolation could not be applied. The same imputation method is provided with TADPOLE-SHARE code.

A complete description of the TADPOLE features with the source table from ADNI is provided at https://github.com/swhustla/pycon2017-alzheimers-hack/blob/master/docs/data_ dictionary.md. The features can be divided into clinical history data, cognitive features, Apolipoprotein E4 gene (APOE4), summary anatomical (MRI, DTI) and metabolic features (PET) computed from images, and cerebrospinal fluid (CSF) biomarkers. These features are typically found in the ADNIMERGE table, which contains key ADNI features in a single table. Additional features from the UCSFFSL, UCSFFSX, BAIPETNMRC, UCBERKELEY, DTIROI, and UPENNBIOMK9 tables are included in the list of TADPOLE features. UCSFFSL and UCSFFSX features are obtained from FreeSurfer segmentations of magnetic resonance imaging (MRI) (L denotes the longitudinal while X denotes the cross-sectional value). The BAI-PETNMRC features are summaries from positron emission tomography (PET) images. UCBERKELEY are features from UC Berkeley Florbetapir F18-AV-45 PET (AV45) and F-AV1451 Brain Tau PET (AV1451) analysis. DTIROI are measures from diffusion tensor imaging (DTI). Finally, UPENNBIOMK9 are features from CSF puncture results. These are biomarkers of proteomic nature. For many TADPOLE participant methods, the clinical diagnosis is considered as a feature.

3 TADPOLE Challenge methods

The exact reproducibility of the best-performing TADPOLE Challenge methods is difficult owing to the lack of both code availability and the scarce details of their implementation. As one of the objectives of this study is to determine the contribution of pure machine learning methods to the accuracy in diagnosis prediction and focus on interpretability, we implemented plain versions of the different systems trained with the underlying machine learning method. In the following, we describe the methods used in our work as well as the factors that distinguish them from the TADPOLE Challenge participant methods.

3.1 Gradient Boosting (XGB)

The winner of the diagnosis forecast in the TADPOLE Challenge was *Frog*, a system built on a gradient boosting machine with XGBoost [27]. Gradient boosting is a model ensemble of individual decision trees that are trained sequentially such that a new tree improves the error of the previous tree ensemble. XGBoost is an optimized distributed gradient boosting library. Its high efficiency is achieved through a parallel tree boosting algorithm that is known to accurately solve data science problems involving billions of examples.

The most relevant hyperparameters of gradient boosting are the learning rate (learning_rate), maximum tree depth (max_depth), number of trees to fit (n_estimators), and L^2 regularization weight (reg_lambda). We performed hyperparameter selection using a randomized search with the RandomizedSearchCV class available in *scikit-learn* with five-fold cross correlation. We found that the default hyperparameters provided a system whose accuracy is close to that of the best-performing model. Therefore, XGBoost has been executed in our study with the default hyperparameter values: learning_rate = 0.3, max_depth = 6, n_estimators = 100, and reg_lambda = 1.

In the TADPOLE one-year follow-up paper [16], the XGBoost system of *Frog* used 70 features from the original data. These features were augmented, yielding a total of 420 features. The selected features included clinical diagnosis, cognitive tests (ADAS-Cog13, CDRSB, MMSE, RAVLT), MRI, FDG-PET, CSF measurements, and APOE status. However, the exact list of features is not available. Augmentation was performed on the basis of the most recent measurement and time, historical highest and lowest measurements and time, and most recent change in measurement. Missing data were filled automatically by XGBoost through inference based on the reduction of the training loss. Prediction was performed using different forecast windows.

We depart from *Frog* system implementation by using all the available features and the longest available forecast window. Augmentation was not considered in this work. Although XGBoost can automatically fill in the missing data, we decided to conduct the same imputation procedure for all the methods in our study.

3.2 Random Forests (RF)

The runner-up of the diagnosis forecast in the TADPOLE Challenge was *ThreeDays*, a random forest (RF) machine. RF is a model ensemble that consists of a large number of decision trees. Each individual tree in the forest performs one class prediction, and the class with the majority of the votes becomes the model prediction. The trees in a forest are uncorrelated models that together produce ensemble predictions that are more accurate than any of the individual predictions. Uncorrelation is the key to a successful ensemble. It is obtained from random selection of different sets of data and different features for each tree.

The most relevant hyperparameters of RF are the method for sampling data points (bootstrap), maximum number of levels in the tree (max_depth), number of features to be considered at every split (max_features), minimum number of samples required at each leaf (min_samples_leaf) and to split a node (min_samples_split), and number of trees in the forest (n_estimators). As with XGB, we performed hyperparameter selection using RandomizedSearchCV. We also found that the default hyperparameters provided the same accuracy as those of the best-performing model. Therefore, RF has been executed in our study with the default hyperparameter values: bootstrap = True, max_depth = None,max_features = auto,min_samples_leaf = 1,min_ samples split = 2, n estimators = 100.

In the TADPOLE one-year follow-up paper [16], the *ThreeDays* system was built with a manual selection of 16 features, namely clinical diagnosis, age, months since baseline, gender, race, marital status, cognitive tests (MMSE, CDRSB, ADAS11, ADAS-Cog13, RAVLT immediate, learning, forgetting, and percent forgetting, FAQ), and APOE status. No imputation was performed in the data, as their RF implementation deals with missing data automatically, by finding optimal splits based on existing data. Two RFs were trained, the first for transitions from CN to AD and the second for transitions from MCI to AD. The authors made the following assumptions in the model: CN individuals always evolve to MCI, MCI individuals evolve to AD, and the diagnosis does not change for AD individuals. The model does not consider valid backward situations in diagnosis. A similar method with a single random forest for all the predictions has been described in [17].

Again, we depart from the *ThreeDays* system implementation using all the available features and a single RF for the diagnosis prediction problem. Our RF implementation did not deal with missing data automatically; hence, the missing data were imputed. We do not consider the diagnosis assumptions, as we can find some counterexamples in the data owing to possible corrections in diagnosis.

3.3 Support Vector Machines (SVM)

The top three diagnosis forecasts in the TADPOLE Challenge are completed with *EMC-EB*, a support vector machine (SVM). Given the number of features, *n*, SVM aims to find a hyperplane in an *n*-dimensional space that best classifies the training data points. The hyperplane maximizes the distance between the points of both classes (maximum margin). The selection of the maximum margin hyperplane as the classification boundary increases the confidence in the algorithm for obtaining correct classifications in unseen data.

The most relevant hyperparameters of SVM are the kernel, the C-parameter, which is a regularization parameter related to the margin size, and γ , which is used to set the size of the kernel. In this work, we used a radial basis function kernel (RBF), C was set to 0.5, and γ was automatically selected depending on the number of features. These hyperparameter values were selected by the authors of *EMC-EB*.

In the TADPOLE one-year follow-up paper [16], the *EMC-EB* system used 200 features from the original data. The features were selected from those showing the largest change over time in subjects who progressed to AD. The file can be found with TADPOLE-SHARE source code (http://tadpole-share.github.io). The selected features included clinical diagnosis, cognitive tests, FreeSurfer cross-sectional MRI volumes, PET and DTI measures, and CSF features. Missing data were imputed using nearest-neighbor interpolation from the subject's earlier timepoints. When this was not possible, the missing data were imputed from the mean values in the training set.

In this work, we only depart from the *EMC-EB* system implementation by using all the available features.

3.4 Benchmark methods

Benchmark methods were provided by the TADPOLE Challenge organizers. The source code was offered to the participants before the deadline as a starting point for making predictions. The evaluation scores obtained using these methods constitute the lower bound of the accuracy expected for the best-performing methods.

3.4.1 Last visit. This method uses clinical diagnosis as the only available feature. The prediction is obtained by assigning 100% probability to the last available diagnosis and 0% to the other diagnoses.

3.4.2 Mixed effects APOE. This method uses the diagnosis, age, ADAS13, and APOE status as features. A mixed-effects model is built using the APOE status as a covariate. The subject age at each visit is selected as the predictor variable, and the predictions are derived from the ADAS13 forecasts using three Gaussian likelihood models for the CN, MCI, and AD status. Then, likelihoods are converted into probabilities by normalization and, the prediction is computed from these probabilities.

3.4.3 SVM. This method uses the clinical diagnosis, age, ADAS13, ventricles, ICV, and APOE status as features. The SVM is implemented with a linear kernel. Missing data are imputed using the average value of the biomarker from past visits of the same subject or the population average if past information is not available. In this work, we also consider an SVM trained with the whole set of 1818 features.

3.5 TADPOLE Challenge metrics for evaluation

The metrics proposed in the TADPOLE Challenge for the evaluation of the systems in the problem of clinical status prognosis were the multi-class area under the receiver operating curve (mAUC) and the overall balanced classification accuracy (BCA). The detailed expressions of these metrics can be found in https://tadpole.grand-challenge.org/Performance_Metrics.

The mAUC is an extension of the classical ROC analysis from binary to multi-class problems. Given $AUC(C_i|C_j)$, the AUC for the classification of a class C_i against C_j , the overall mAUC is given by

$$mAUC = \frac{2}{C(C-1)} \sum_{i=2}^{C} \sum_{j=1}^{i} \frac{AUC(C_i|C_j) + AUC(C_j|C_i)}{2},$$
(1)

where *C* is the number of considered classes. Intuitively, the mAUC measures the degree to which the model is capable of correctly distinguishing between classes; a higher degree is desirable.

The overall BCA is given by the mean of the balanced accuracies for each class:

$$BCA = \frac{1}{C} \sum_{i=1}^{C} BCA_i,$$
(2)

where

$$BCA_{i} = \frac{1}{2} \left(\frac{TP}{TP + FN} + \frac{TN}{TN + FP} \right), \tag{3}$$

and TP, FP, TN, and FN represent the number of true positives, false positives, true negatives, and false negatives, respectively, for the class C_i . Intuitively, the BCA provides a balanced assessment of true positives/negatives and false positives/negatives obtained by the model.

4 Explainable Al

Explainable AI (XAI) is a recent subfield of AI that aims to provide explanations of general machine learning algorithms while focusing on the most complex families of methods that have been traditionally considered as black boxes [18, 20]. The objective of XAI is to leave out the trade-off existing between accuracy and explainability and provide both powerful and

explainable systems with arguments for increasing the confidence in the output of the algorithms.

Explainability technically highlights the relevant parts of a system that contribute toward model accuracy during training or toward a specific correct or incorrect prediction for a particular observation during testing or evaluation [28]. The concept has been recently linked with causability, which is defined as the extent to which an explanation to a user achieves a specified level of effective, efficient, and satisfactory causal understanding in the context of use [29, 30]. Causability provides criteria for measuring and ensuring the quality of explanations. Explainability linked with causability improves causal understanding; thus, AI methods and explanations become trustable and actionable, i.e., the actions that should be taken become clear for decision makers [30]. Thus, explainability is important first step toward fully actionable and interpretable AI systems.

The different techniques in XAI facilitate the establishment of the importance given to the different features by the system, assessing how a particular feature affects model predictions, or which feature values favor or hinder correct or incorrect predictions. All this information can help ensure whether only significant features coherent with current knowledge are used to infer the output for the application of interest. In this work, we focus on the built-in importance scores of interpretable machine learning methods and SHAP.

4.1 Built-in importance scores

XGB and RF are ensembles of decision trees; therefore, the models can provide estimates of the importance given to different features. In XGB, the importance measures how valuable each feature is in the construction of the boosted decision trees within the model. The more an attribute is used to make key decisions within the trees, the higher is its relative importance. The importance is measured from the amount by which each feature split point improves the performance measure (gain). Another measure of importance is the cover, which summarizes the second-order gradient (Hessian) of the loss function for all the predictions of a point in a tree. Accordingly, XGB provides five ways for computing feature importance:

- 1. weight: directly measures the number of times a feature is used to split the data across all trees
- 2. gain: average gain across all splits in which a feature is used
- 3. cover: average coverage across all splits in which a feature is used
- 4. tgain: total gain across all splits in which a feature is used
- 5. tcover: total coverage across all splits in which a feature is used

The gain and cover are computed from the quotients of the total gain and total cover with the weight. These metrics are usually recommended for quantifying importance. However, it should be noted that the weight value for a feature depends on the range of different values taken by that feature. Therefore, there may be important variables for the model with a low weight because the range of different values is low (e.g., gender). In this situation, the total gain or the total cover value is preferred. This is the case for our study.

In RF, feature importance is computed from the mean and standard deviation of the accumulation of the impurity decrease within each tree (Gini importance). As with the XGB importance metrics, the feature importance metrics here may be misleading for features with a low number of different values. For kernel-based SVM, there is no built-in way of establishing feature importance.

4.2 SHapley Additive exPlanations

SHapley Additive exPlanations (SHAP) has attracted increasing attention in the field of XAI (https://shap.readthedocs.io). The core idea of SHAP is to transfer ideas from cooperative game theory to the attribute feature importance of a model output given an input [31]. SHAP values represent the distribution of the contributions toward game success or failure amongst all the players working in cooperation. In the context of explainability of machine learning methods, SHAP values represent the change in each feature in the expected model prediction under conditioning on that feature. The explainability from SHAP is achieved through different representations derived from the SHAP values. In this work, we use the mean absolute values of the SHAP values obtained for each class and the violin plots of the SHAP value impact on the estimation of the probabilities for each class.

S1 Fig in S1 File shows typical examples of the SHAP value representation for a given problem (survival classification in the Titanic dataset). The horizontal bar plot shows the mean absolute SHAP values of the most relevant features for survival classification. The plot shows that gender is the most relevant feature for the correct classification of survivals, followed by the title code of the person and the class. These results are coherent with our knowledge of the Titanic tragedy. Therefore, the underlying machine learning method can generate a model that is aware of the outcome of survival owing to the human decisions made during the disaster.

The violin plot in Fig 1 represents the impact of the feature values on the probability computed for the survival class by the model. The color code indicates the feature values for different test samples, and it is useful to relate whether the high or low probabilities computed by the model are favored by given feature values. The color bar ranges from blue tones for low values to red tones for high values. Thus, the model favors high values of the probability of survival for high values of gender (i.e., female). From the class feature, we can see that the model favors high values of the probability of survival for low values of class (i.e., first class). In our study, we use similar reasoning to establish the relevance of the features in the identification of different classes (AD, MCI, or CN).

For XGB and RF, the SHAP values are obtained from tree explainers, a fast implementation of SHAP for tree-based methods [32]. For SVM, the SHAP values are obtained from kernel explainers, an agnostic algorithm that consequently entails a higher computational cost.

5 Results

5.1 Performance study

5.1.1 Performance results. Table 2 (left) summarizes the performance of the TADPOLE Challenge methods of interest in this work on the evaluation set (D4). Table 2 (right, top) shows the performance of the baseline methods (last Visit, mixed effects, and linear SVM). These results are the same as those reported at https://tadpole.grand-challenge.org/Results and [16]. Table 2 (right, bottom) also summarizes the performance of the baseline methods trained with our data on the test (D2) and evaluation (D4) sets. These results were obtained by integrating the TADPOLE-SHARE code (https://tadpole-share.github.io) into our code. The results obtained on D4 were close to the TADPOLE Challenge results. Therefore, our choice of data preprocessing provides a similar performance on the benchmark methods.

Table 3 summarizes the performance of the methods considered in this work on the test (D2) and evaluation (D4) sets. We performed an ablation study on feature set selection. Therefore, the experiments were conducted by removing different sets of features, namely, the diagnosis (-D), diagnosis and cognitive features of TADPOLE used in ADNI to establish diagnosis



 $XGB-D_{ADNI}$

Fig 1. XGB built-in importance. Total gain scores obtained by different XGB models. Scores averaged over 10 random experiments for the 100% training set. Long feature names have been trimmed for better legibility. The bars are colored by feature type (diagnosis, clinical, cognitive, MRI, PET, DTI, APOE4, and proteomic biomarkers).

https://doi.org/10.1371/journal.pone.0264695.g001

 $(-D-D_{ADNI})$, diagnosis and cognitive features (-D-C), and diagnosis and image-based features (-D-I). The models were built with balanced training sets by selecting the 100% of the underrepresented class (AD). In addition, S1-S3 Tables in S1 File compare the performance of the models built with different sample sizes of the under-represented class (100, 50, and 25%, respectively).

5.1.2 TADPOLE rank preservation. Comparing the performance metrics obtained with the TADPOLE results in Table 2 with those of our models in Table 3, we find that *Frog*,

	Evaluation set (D4)				
	rank			mAUC	BCA
Frog	1		0.931		0.849
ThreeDays	2		0.921		0.823
EMC-EB	3		0.907		0.805
	Evaluation set (D4)				
	rank			mAUC	BCA
SVM	30			0.836	0.764
Mixed Effects APOE	35		0.822		0.749
Last visit	44-45	44-45		0.774	0.792
	Test set (D2)			Evaluation set (D4)	
	mAUC	BC	A	mAUC	BCA
SVM, 1818 feat.	0.877	0.76	55	0.828	0.744
SVM, 6 feat.	1.000	1.00	00	0.855	0.792
Last visit	1.000	1.00	00	0.774	0.792

Table 2. Performance reported in the TADPOLE Challenge on D4. Performance obtained with the benchmark methods in D2 and D4.

Left: performance reported in the TADPOLE Challenge on the evaluation set (D4) for the methods considered in this work (results shown in [16], Table 2). Right, top: performance of the benchmark methods on the evaluation set (D4) reported in the TADPOLE Challenge (results shown in [16], Table 2). Right, bottom: performance of the benchmark methods on the test (D2) and evaluation (D4) sets obtained with our data.

https://doi.org/10.1371/journal.pone.0264695.t002

Table 3. Performance obtained with the methods considered in this work on the test (D2) and evaluation (D4) sets.

	Test set (D2)		Evaluation set (D4)	
	mAUC	BCA	mAUC	BCA
XGB, 100%	0.998 (± 0.000)	0.982 (± 0.006)	0.909 (± 0.004)	0.804 (± 0.003)
XGB-D, 100%	0.945 (± 0.001)	0.863 (± 0.009)	0.897 (± 0.004)	0.788 (± 0.010)
XGB-D _{ADNI} , 100%	0.920 (± 0.005)	0.833 (± 0.005)	0.881 (± 0.005)	0.795 (± 0.008)
XGB-D-C, 100%	0.791 (± 0.008)	0.680 (± 0.013)	0.732 (± 0.011)	0.646 (± 0.016)
XGB-D-I, 100%	0.942 (± 0.004)	0.873 (± 0.003)	0.909 (± 0.006)	0.820 (± 0.009)
RF, 100%	0.976 (± 0.003)	0.908 (± 0.008)	0.894 (± 0.004)	$0.804 (\pm 0.008)$
RF-D, 100%	0.933 (± 0.003)	0.832 (± 0.008)	0.880 (± 0.005)	0.776 (± 0.012)
RF-D _{ADNI} , 100%	0.904 (± 0.005)	0.795 (± 0.010)	0.864 (± 0.006)	0.759 (± 0.008)
RF-D-C, 100%	0.774 (± 0.006)	0.671 (± 0.013)	0.721 (± 0.010)	0.625 (± 0.016)
RF-D-I, 100%	0.945 (± 0.003)	0.878 (± 0.011)	0.900 (± 0.004)	0.803 (± 0.008)
SVM, 100%	0.925 (± 0.002)	0.815 (± 0.006)	0.845 (± 0.003)	0.757 (± 0.003)
SVM-D, 100%	0.909 (± 0.002)	0.785 (± 0.007)	0.839 (± 0.003)	0.739 (± 0.005)
SVM-D _{ADNI} , 100%	0.893 (± 0.002)	$0.784 (\pm 0.008)$	0.830 (± 0.003)	0.732 (± 0.007)
SVM-D-C, 100%	0.773 (± 0.006)	0.656 (± 0.011)	0.721 (± 0.006)	0.619 (± 0.011)
SVM-D-I, 100%	0.917 (± 0.002)	0.816 (± 0.009)	0.877 (± 0.002)	0.777 (± 0.093)

Models built with different feature sets: whole, without clinical diagnosis (-D), without clinical diagnosis and ADNI features for diagnosis (-D-D_{ADNI}), without clinical diagnosis and cognitive features (-D-C), and without clinical diagnosis and image features (-D-I). The training sets are balanced and use 100% of the under-represented class (AD). For each method configuration, the mean and standard deviation of the mAUC and BCA obtained over 10 experiments are shown.

https://doi.org/10.1371/journal.pone.0264695.t003

ThreeDays, and *EMC-EB* outperform our methods (on D4, mAUC of 0.931 vs. 0.909, 0.921 vs. 0.894, and 0.907 vs. 0.845, respectively). Frog differs from our XGB system in terms of the training set, set of features, imputation, use of augmentation, and use of different temporal windows. *ThreeDays* differs from our RF system in terms of the training set, set of features, imputation, and use of forest ensembles in solving pairwise problems. *EMC-EB* differs from our SVM system in terms of the training set and set of features. Thus, the performance improvement achieved by the winners depends more on the data used, feature selection, and mentioned *ad hoc* methodological modifications than on the underlying machine learning method.

Nevertheless, the results in Table 3 for the models trained with all the features show that the ranks obtained in D2 and D4 by our systems are consistent with the TADPOLE Challenge podium. XGB obtains the best mAUC and BCA metric values, closely followed by RF, while the metrics obtained by SVM are considerably lower. Therefore, either XGB or RF seems to be the most competitive method for the problem of AD/MCI/CN identification. RF was also the best-performing method in the exhaustive study presented for a different application in [33].

5.1.3 Feature set selection and performance. In D2, the influence of feature set selection on the performance metrics is considerable. The best results are obtained for the feature set including the clinical diagnosis. The feature sets excluding the diagnosis (-D) degrade the performance of the methods considerably even though the clinical diagnosis is partially obtained from the D_{ADNI} features. For example, the mAUC for XGB decreases from 0.998 to 0.945. For RF, the mAUC decreases from 0.976 to 0.933. For SVM, the mAUC decreases from 0.925 to 0.909. A further decrease in the feature set with the diagnosis and D_{ADNI} features further degrades the performance.

The methods with feature sets excluding diagnosis and image features (-D-I, i.e., including cognitive features) perform consistently better than those with feature sets excluding diagnosis and cognitive features (-D-C, i.e., including image features). For RF and SVM, excluding diagnosis and image features leads to slight improvement in performance compared to excluding diagnosis. For XGB, both performances are nearly identical.

In D4, the overall best results are obtained for the feature set excluding clinical diagnosis and image features (-D-I). The feature set including all the features achieves the second position, and removing the diagnosis and subsequent D_{ADNI} features has the effect of slightly degrading the performance. In this case, the loss in performance with the -D-C feature set is considerable, i.e., it is lower than the last visit benchmark performance.

5.1.4 Robustness to sample size selection. The robustness of the results with respect to the selection of the training set size was assessed through 10 different experiments for each method and feature set. For each experiment, the data were selected randomly, except for the under-represented class in the 100% experiment. Recall that S1-S3 Tables in <u>S1 File</u> list the mean and standard deviation of the mAUC and BCA obtained by the methods in D2 and D4.

In general, the metrics show a small standard deviation in both D2 and D4 with variations in the third decimal position in the majority of the cases. Therefore, all the models obtain consistent results for the same configuration of sample and feature selection. The models trained with 25% of the samples of the under-represented class obtain metrics slightly smaller but close to those of the models trained with the whole under-represented class. Therefore, the performance achieved by the methods can be considered robust to these changes in data sampling. Now, we proceed with the interpretability study for the 100% sample sets.

5.2 Interpretability study

5.2.1 XGB built-in importance scores. Fig 1 shows the 25 most important features according to the total gain importance of XGB averaged over 10 random experiments for the

100% training sets. Recall that the legend of the feature names can be found at https://github. com/swhustla/pycon2017-alzheimers-hack/blob/master/docs/data_dictionary.md. The diagnosis is scored as the most important feature. The following features show a considerable total gain reduction.

The most relevant features include the diagnosis, age of each patient at baseline (Years_bl), and several measurements from cognitive tests (CDRSB, FAQ, ADAS13, RAVLT, CDRSB_bl, DX_bl, ADAS11, and ECog). Then, a number of measurements from anatomical structures obtained from the FreeSurfer cross-sectional pipeline (UCSFFSX) arise, interleaved with the ADNIMERGE image and other cognitive measurements. The anatomical structures include the amygdala (ST17SV), hippocampus (ST29SV), inferior and middle temporal regions (ST32TA, ST40TA), entorhinal region (ST83CV), and cuneus (ST23SA).

When the diagnosis is removed from the XGB models, CDRSB, the diagnosis at baseline DX_bl, and CDRSB_bl occupy the first, second, and third positions, respectively. The previously mentioned cognitive features increase their relative importance. CDRSB and MMSE, used in ADNI for establishing diagnosis, occupy the top positions. Different UCSFFSX, UCSFFSL, UCBERKELEY, and DTI features can be now found in the last top 25 positions.

When removing the features used in ADNI for establishing diagnosis, FAQ and ADAS13 occupy the top positions. In this case, the UCSFFSX features are mostly replaced by the BAI–PET features. These features are now highly interleaved with cognitive data. The BAIPET features include FDG measurements from the posterior cingulate (FEMPC), different AV45 measurements from the temporal region (AVEMTEMP, AVNATEMP, TMPINFL), and the hypometabolic index (HCI). We can easily find studies in the literature on the role of PET imaging of these regions in the characterization of AD [34, 35]. The hypometabolic index has been also identified as a biomarker useful for the characterization of the disease [36].

When removing all cognitive data, the UCSFFSX and BAIPET features mostly occupy the top 25 positions. Some features of relevance for the disease, such as FDG-PET, the hippocampus volume, or APOE4, can be seen in the top 25 set. The features from the posterior cingulate, precuneus, temporal region, and hypometabolic index still appear in similar positions. The importance is distributed over a wide number of features. From the UPENNBIOMK9 biomarkers, ABETA and PTAU are in the top 25 set.

Removing image data makes cognitive features prevail in importance, and they occupy the top 25 positions. The most important features are the features used for diagnosis in ADNI. Then, the ADAS and RAVLT features follow in importance. Finally, the ECog features occupy the last of the top 25 positions. In addition, the three UPENNBIOMK9 features (ABETA, TAU, PTAU) are now in the top 25 positions.

Fig 2 shows the 25 most important features according to the total cover importance of XGB averaged over 10 random experiments for the 100% training sets. In general, we can see discrepancies among the features selected according to the total gain and total cover although there is agreement in the features selected in the first positions.

5.2.2 RF built-in importance scores. Fig 3 shows the 25 most important features according to the built-in importance of RF averaged over 10 random experiments for the 100% training sets. The most striking difference from the XGB results is that the importance of RF is distributed over a greater number of features. The diagnosis is again the most important feature, mostly followed by cognitive features (CDRSB, CDRSB_bl, FAQ, MMSE, ADASs, etc.) Only one FreeSurfer UCSFFSX feature from the entorhinal region (ST24TA) occupies one of the last top 25 positions.

Removing the diagnosis and the features used in ADNI for diagnosis results in cognitive features similar to XGB occupying the top positions. However, RF seems to be more dependent on the information given by cognitive data compared to XGB.





Upon removing all the cognitive data, the UCSFFSX features mostly occupy the top 25 positions. Some features of relevance for the disease, such as the hippocampus (ST29SV, Hippocampus, ST88SV), FDG-PET (FDG), entorhinal volume (ST24TA, ST83TA, Entorhinal), and middle and inferior temporal regions (ST40TA, ST32TA), can be seen among these features. Removing image features results in a configuration of top cognitive features nearly identical to XGB. Here, the UPENNBIOMK9 features do not appear in the top 25 positions.



Fig 3. RF built-in importance. Importance scores obtained by different RF models. Scores averaged over 10 random experiments for the 100% training set. Long feature names have been trimmed for better legibility. The bars are colored by feature type (diagnosis, clinical, cognitive, MRI, PET, DTI, APOE4, and proteomic biomarkers).

5.2.3 SHAP-based explainability using bar plots. Figs 4–6 show the 25 most important features according to the mean absolute SHAP values in one of the experiments for the 100% training sets and the models including the whole feature set. The diagnosis is recognized by SHAP as the most important feature. The relative importance compared with subsequent features is remarkably greater for XGB and SVM than for RF.

For XGB, the most relevant features include the diagnosis, age of each patient at baseline (Years bl), some features characterizing the cognitive condition (CDRSB, ADAS, FAQ,





https://doi.org/10.1371/journal.pone.0264695.g004









https://doi.org/10.1371/journal.pone.0264695.g006

RAVLT, CDRSB_bl, and ECog), and measurements from FreeSurfer cross-sectional segmentations (UCSFFSX). These measurements refer to the volume, surface area, or cortical thickness of regions related to AD, such as the hippocampus (ST29SV), entorhinal region (ST24CV, ST83CV), inferior temporal region (ST32S, ST32T), and temporal pole (ST60).

SHAP feature ranking in the first positions is more coherent with the XGB total gain compared to the total cover built-in importance results. In addition, SHAP provides information regarding the importance of each feature to the different classes. For the identification of CN individuals, the diagnosis is mainly the most relevant feature. For the diagnosis of AD individuals, the diagnosis, CDRSB, Years_bl, ADAS13, and FAQ are the most relevant features. Thus, the relevance of CDRSB_bl and DX_bl in the diagnosis of AD is quite low. As these features are not useful for the correct classification of converters, it seems that the system can identify the diagnosis and CDRSB as preferable features for AD diagnosis. For the identification of MCI individuals, the importance of diagnosis is much greater than the importance of other features.

For RF, the most relevant features are all cognitive, including CDRSB_b1, CDRSB, DX_b1, FAQ, ECog, ADASs, MMSE, and RAVLT. The only image-based feature is the cortical thickness average of the left entorhinal region (ST24T). As with XGB, the diagnosis is more relevant for the identification of CN and MCI individuals. In contrast to the XGB results, the importance of the features is much more distributed among the top 25 features for the CN and AD classes. The built-in importance results showed a similar set of features in similar positions.

For SVM, the diagnosis and DX_bl are the most relevant features, followed by FAQ, CDRSB, and CDRSB_bl. Subsequently, a number of image-based features are found, including the BAIPET, FFSL, AV45, FFSX, and DTI features. These features refer to regions such as the posterior cingulate gyrus (FEMPC2), temporal area (AVETEMP2), parahippocampal



Fig 7. XGB, RF, and SVM explainability. The violin plots show the SHAP value impact on the estimation of the probabilities for each class of the XGB, RF, and SVM models. Models trained with all the features. Long feature names have been trimmed for better legibility.

region (ST44SA), pallidum (ST24SV), parietal and precentral areas (ST90TS, ST110TA), fornix (FAFXSTR), superior frontal area (ST56CV), isthmus (ST34SA), and superior temporal area (ST58CV). Some more cognitive features such as MMSE and RAVLT_immediate can be found in baseline-current visit pairs. APOE4 is the last feature in the top 25 positions. Thus, the AD class is mostly explained by the diagnosis, FAQ, and CDRSB, and, to a lesser extent, by MMSE.

5.2.4 SHAP-based explainability using violin plots. Fig 7 shows the impact of each feature on the different model outputs for each class. The impact is represented by violin plots, as is customary in SHAP analysis. Comparing the distribution of the violin plots of RF and XGB, we can see that the contribution of the first features to the probabilities is polarized in two modes for a greater number of features. Thus, for RF, the values of all these features are more strongly considered in the classification. For SVM, the distribution of the violin plots shows a single mode for all features.

The impact representation of the diagnosis feature suggests that low diagnosis values favor high probabilities for the CN class (p(CN)), and low probabilities for the MCI and AD class (p(MCI) and p(AD)). High diagnosis values favor low p(CN) and high p(AD). Mean diagnosis

values favor high p(MCI). As the labels assigned to each class are 0 for CN, 1 for MCI, and 2 for AD, all the models can capture the relationship between labels and future status.

The violin plots of Years_bl for XGB indicate that the system considers the information of the age when the patient is included in the study. The greater the age, the greater is the p (*AD*). This is consistent with the knowledge that AD is linked with age. The model also considers the months from baseline (M) as important.

Regarding the CDRSB_bl and CDRSB features, the lower the values, the greater is p(CN) and viceversa for p(AD). High values of p(MCI) are favored by mean values of CDRSB. The scores reflect the Hughes Clinical Dementia Rating (CDR), where a CDR of zero indicates that the subject is healthy and gradually increases until values of 3, indicating that the subject suffers from a serious cognitive decline.

Regarding cognitive scores, a low value in FAQ, ADAS13, and ADAS11 and a high value in RAVLTs favor high p(CN), and viceversa for AD. It has been reported that higher Adas-Cog and lower RAVLT scores suggest greater impairment [37]. Regarding MRI features, medium to high volume values favor high values of p(CN). This indicates that the systems can associate atrophy with high values of p(AD) and vice versa for p(CN).

In summary, the violin representation of the SHAP values shows that the three models can establish a relationship between feature values and diagnosis probability consistent with clinical knowledge for the top 25 features. The distribution of the SHAP value impact tends to be bi-modal for the most important features of the best-performing methods.

5.2.5 Removing diagnosis, cognitive, and image features. Fig 8 shows the reassignment of importance to the remaining features of the XGB, RF, and SVM models after progressively removing the diagnosis (-D), the features used by ADNI in diagnosis (-D- D_{ADNI}), cognitive (-D-C), and image (-D-I), features. The different systems rebalance the importance given to the features, and the contribution becomes less biased to a single prevalent feature.

After removing diagnosis, the features used in ADNI for the diagnosis occupy top positions, as with the built-in importance study. CDRSBs features are found to be much more relevant than MMSE. For XGB, image-based features from the hippocampus and entorhinal region prevail. For SVM, most image-based features prevail.

After removing the diagnosis and D_{ADNI} features for XGB and RF, FAQs the features become the most important, followed by cognitive features such as ADAS, RAVLT, and ECog. For XGB, Years_bl still occupies the top positions. For SVM, the behavior is slightly different from XGB and RF. FAQ still tends to occupy top positions. However, image-based features such as BAIPET, FFSL, and FFSX increase their relevance to the model. These features relate to regions such as the posterior cingulate gyrus for FDG-PET and the temporal region for AV45-PET. The unknown region (ST123TA) also appears as relevant for SVM.

After removing all the cognitive features, the system again rebalances the importance given to the features, and image-based features occupy higher positions. Some MRI and PET measurements such as FDG, the hippocampus volume, FDG_bl, and the entorhinal volume now increases their importance. Both XGB and RF agreed on the importance of the anatomical features related to the entorhinal region, hippocampus, middle temporal region, and amygdala. In addition, both methods agreed on the importance of the hypometabolic index, FDG measurements of the posterior cingulate gyrus, and AV45 measurements of the temporal region. For SVM, the FDG-PET measurements of the posterior cingulate gyrus and the AV45-PET measurements of the temporal region were the most important features. APOE4 reaches the top 5 positions in importance. After removing image-based features, cognitive features mostly occupy the top 25 features. For XGB, the TAU and ABETA biomarkers are in the top 25 positions.



Fig 8. Mean absolute SHAP values computed from the XGB, RF, and SVM systems with feature ablation. First and second rows: models trained after removing diagnosis (-D) and features used in ADNI for diagnosis (-D-D_{ADNI}). Third and fourth rows: models trained after removing cognitive (-D-C) and image-based (-D-I) features. Note that the label coloring is not consistent among experiments owing to rigid SHAP implementation. Long feature names have been trimmed for better legibility.

The supplementary material shows the violin plots of the impact of the features on the model outputs for the different feature sets (S2-S4 Figs in <u>S1 File</u>).

5.3 Robustness to sample selection

Figs 9-11 show the frequency of the top 25 features over 10 experiments for the 100% training set and the different feature sets. The results can be found in the csv files of the supplementary material: S1-S3 Data, respectively. For XGB and the whole feature set, around 50% of the top





25 features are consistently shown in the top 25 positions. These features include the diagnosis, Years_bl, and cognitive features. The most frequent UCSFFSX features are volume and area measurements from the hippocampus, entorhinal region, superior temporal sulcus, and inferior temporal region. Upon subsequently removing the diagnosis, ADNI, cognitive, and image-based features, the number of features consistently increases. For the -D and -D-D_{ADNI} sets, the most consistent features are mostly cognitive. For the -D-C feature set, apart from





measurements related to the entorhinal region, hippocampus, and middle temporal region, the most frequent features in the top 25 positions are FDG, PET AV45 measurements in the temporal region, the hypometabolic convergence index, and APOE4, ABETA, and PTAU values. For the -D-I feature set, the top 25 most selected features are CDRSB, FAQ, RAVLT, ADAS scorings, etc., and not cognitive features such as TAU, ABETA, and Years bl.



Fig 11. Robustness of SHAP importance for SVM. Frequency of the top 25 features over 10 random experiments for the 100% training set. The bars are colored by feature type (diagnosis, clinical, cognitive, MRI, PET, DTI, APOE4, and proteomic biomarkers).

For RF, the number of features selected in the top 25 positions is more robust and focused on cognitive features. For the -D-C feature set, the measurements from the entorhinal region, hippocampus, inferior lateral ventricle, and middle temporal region can be consistently seen in the top 25 positions. For the -D-I feature set, the consistency reaches nearly 100% of the features in the top 25 positions with mostly cognitive features.

For SVM, only 25% of the top 25 features are consistently shown in the top 25 positions. These features include the diagnosis, diagnosis at baseline, BAIPET, and cognitive features. For the whole, -D, and $-D_{ADNI}$ feature sets, feature importance is quite inconsistent. For the -D-C and -D-I feature sets, feature importance is much more robust. For the -D-C feature set, the BAIPET features are mostly selected together with APOE4 and ABETA UPENNBIOMK9. For the -D-I feature set, the most relevant cognitive features throughout our interpretability study are mostly selected in the top 25 feature set.

6 Discussion of our interpretability study in the context of the state of the art

6.1 TADPOLE Challenge feature importance

According to the global study of feature importance performed in TADPOLE, ABETA from the UPENNBIOMK9 features, image features for DTI, and APOE status contributed toward the best accuracy. The results obtained in our interpretability study did not corroborate these findings for the top three methods. ABETA biomarker only appeared among the best important features in the XGB-D-C, XGB-D-I, RF-C, and SVM-D-I models. We observed that the UPENNBIOMK9 features are considered as relevant for the models obtained with the -D-C or the -D-I sets. Therefore, the UPENNBIOMK9 features may be of importance because many of the TADPOLE Challenge methods used partial sets of features. In addition, DTI features hardly appeared in the built-in feature importance plots. APOE4 was only relevant when XGB was deprived of from cognitive information or SVM was deprived of from ADNI diagnosis, cognitive, or image-based features. The authors of *Frog* reported that the XGB built-in importance scores suggested that the MRI features play a greater role in models trained with long forecast windows. This finding was also obtained from our XGB model, where 10 out of 25 features belonged to the UCSFFSX family.

Meanwhile, the TADPOLE Challenge results indicated that image features from tau PET (AV1451), amyloid PET (AV45), and FDG-PET (FDG) contributed toward the worst accuracy. However, our results showed that the FDG feature was the most important feature in the RF-D-C model and among the most important features in XGB-D-C and SVM-D-C. In addition, FDG_b1 was also present in the top 25 feature set. AV45 appeared frequently in the top 25 set of the RF-C model. Some FDG and AV45 PET features were relevant in models with the -D-C feature set.

The feature selection study performed in [17], the closest journal version of *ThreeDays*, identified the diagnosis, RID (patient identifier), AGE, MIDTEM (middle temporal gyrus), CDRSB, GENDER, FAQ, APOE, and MMSE as the most relevant features for accurate identification of the AD/MCI/CN subjects with RF. Accordingly, our study has corroborated the importance of the diagnosis, CDRSB, FAQ, and MMSE.

The ablation study in [23], the journal version of *CBIL*, highlighted the importance of the diagnosis and CDRSB features in high-performance systems. The authors observed significant drops in the evaluation metrics upon removing these features from their recurrent neural network (RNN) model. Our study also involved this loss in accuracy for the models with the -D and -D-* feature sets.

6.2 Feature importance outside the TADPOLE Challenge

6.2.1 SHAP in RF-based multilayer multimodal detection and prediction model. Outside the TADPOLE Challenge framework, El–Sappagh et al. [24] recently proposed the use of SHAP values in a RF-based multilayer multimodal detection and prediction model. The detection layer performs the classical classification into CN, MCI, and AD. The authors reduced the feature set to a total of 28 baseline features and the results were obtained with a single model optimized with classical cross-validation. Their feature set did not include the diagnosis, as it is undesirable to include the label as a feature for the classification problem. In addition, some features not present in the TADPOLE Challenge were used in the generation of the model.

The study identified CDRSB as the most influential feature for CN and MCI, followed by MMSE for AD. From the TADPOLE Challenge features used in that study, the model recognized cognitive and UPENNBIOMK9 features as the most relevant features. SHAP violin plots for AD showed MOCA, FAQ, ADAS13, and ADAS11 in the top relevant positions. Subsequently, ABETA, PTAU, and TAU were also shown as influential. RAVLT_immediate was also present in the relevance plot. Reasoning regarding feature values and clinical knowledge showed that the model correctly favored the probability of the different classes. For example, high values of CDRSB favored p(AD) whereas low values of CDRSB and high values of MOCA favored p(CN).

In our RF models, cognitive features prevailed in the top positions. In the -D-I model, whose feature configuration is probably the closest to that of El-Sappagh's model, CDRSB was also the most influential feature and FAQ, ADAS13, MMSE, and ADAS11 were shown in the top positions. The violin plots showed similar behavior in favoring the probabilities of the different classes. However, MOCA was found at the 25th position and UPENNBIOMK9s were found beyond the top 25 set. As seen in our ablation study, the importance given by the models to the different features strongly depends on the feature set used to generate the model. It may be possible that the features not present in the TADPOLE Challenge combined with feature selection are responsible for the rise of MOCA and UPENNBIOMK9 as relevant features for El-Sappagh's model.

6.2.2 Systematic, quantitative, and critical review of machine learning in predicting the progression of MCI. The exhaustive review in [38] showed that including cognitive and FDG features significantly improved the predictive performance of the methods compared to not including them. Other modalities, especially MRI-based features, did not show a significant effect. The authors argued that the good performance of cognitive assessments cast doubts on the wide use of imaging for predicting the progression of AD and suggested exploring further fine domain-specific cognitive evaluations.

Our study corroborated the importance given by the different models to the cognitive features. In fact, RF mostly relied on cognitive information. In addition, FDG was identified as a relevant feature when cognitive information is substantially missing. Although the use of cognitive features seems to be critical for an accurate and trustable decision support system in clinical practice, it should be noted that cognitive features are useful only once the patients have started to show some cognitive decline. According to the results of our study, we believe that different sets of features can be used depending on the stage of the disease for accurate computer-aided diagnosis systems. For early diagnosis, we believe that one should discard cognitive information and further study MRI, PET, DTI, biomarkers, and genetic features. Our study has shown that XGB and RF can identify the relevant information according to clinical knowledge for different sets of features while SVM is less accurate and seems to be less reliable. We believe that SHAP values may provide suitable criteria for feature selection.

6.3 Diagnosis as a feature

The clinical diagnosis has been identified in our study as the most relevant feature for all the models. The difference in importance with the following features is considerable for all the methods. In [17] and [23], the relevance of the diagnosis to the accuracy of the prediction for their random forests and RNN systems has been identified.

Table 4 lists the top 25 TADPOLE Challenge methods and indicates whether they used diagnosis in the feature set. We can see that the best-performing methods consistently used the diagnosis. The first two methods that did not use diagnosis were *EMC1-Std* (disease progression model combined with SVM, mAUC = 0.898) and *CBIL* (RNN, mAUC = 0.897). These methods ranked eighth and ninth, respectively. Although the diagnosis was explicitly included in the feature set of the majority of the best-performing TADPOLE methods, we believe that the inclusion of such a feature in a prognostic model may be questionable; moreover, the diagnosis values from the previous visits provide the values of the predicted labels in the training set. Methods not including the diagnosis in their feature set and achieving an accuracy close to 0.90 should be reconsidered as potential well-performing methods for the AD/MCI/CN identification problem.

7 Discussion and conclusions

We investigated the performance and interpretability of the three best-performing methods in the TADPOLE Challenge for the prognosis of clinical AD status. We used the same data

Method	Rank	Use diagnosis
Frog	1	Yes
ThreeDays	2	Yes
EMC-EB	3	Yes
Apocalypse	7	Yes
GlassFrog-Average	5	Yes
GlassFrog-SM	5	Yes
GlassFrog-LCMEM-HDR	5	Yes
EMC1-Std	8	No
CBIL	9	No
CN2L-RandomForest	10	Yes
EMC1-Custom	11	No
BGU-LSTM	12	No
DIKU-GeneralisedLog-Custom	13	No
DIKU-GeneralisedLog-Std	14	No
ARAMIS-Pascal	15	No
VikingAI-Sigmoid	16	Yes
Tohka-Ciszek-RandomForestLin	17	Yes
IBM-OZ-Res	18	Yes
BORREGOTECMTY	19	Yes
VikingAI-Logistic	20	Yes
lmaUCL-Std	21	Yes
lmaUCL-Covariates	22	Yes
Chen-MCW-Stratify	23	No
AlgosForGood	24	No
lmaUCL-halfD1	26	Yes

Table 4. TADPOLE Challenge top 25 methods and use of diagnosis.

https://doi.org/10.1371/journal.pone.0264695.t004

processing choices for fair comparison of the three methods. Interpretability was studied by combining feature ablation with feature importance assessment. We investigated the built-in feature importance metrics, when available, and we computed SHAP values from the models. SHAP is a recent tool for XAI that allows robust study of per-class feature significance and establishes the impact of each feature in the output probabilities. It has been shown to be a useful tool for answering key questions in terms of increasing or decreasing our confidence in conventional machine learning methods that occupy the top positions in the TADPOLE Challenge.

Our models preserved the top three ranks achieved in the TADPOLE Challenge, i.e., gradient booster ranked in the first position, random forest in the second one, and support vector machines closed the top three. However, the performance achieved on the evaluation set D4 was consistently lower than that reported by the TADPOLE Challenge results. We believe that these differences are due to the data used, feature selection, and methodological modifications to the standard machine learning method. For example, we believe that the accuracy of *Frog* is due to the combination of gradient booster with feature selection and data augmentation. *ThreeDays* is a model ensemble of two models specialized in binary classification problems with aggressive selection of features that may be the key to improved accuracy.

The accuracy of the models was robust to the selection of the training data. The inclusion of 1818 features did not degrade the performance of the models. It seems that once the diagnosis is removed, a number of cognitive features are required for acceptable performance. Otherwise, the performance of the system is considerably degraded. This is consistent with results shown by the state of the art. For example, the CadDementia Challenge winner [13] obtained a limited AUC of 0.788 in the problem of AD/MCI/CN classification from MRI image-based features. Spasov et al. [39] obtained an AUC of 0.925 by combining cognitive and structural MRI images for the problem of sMCI/pMCI identification. The AUC decreased to 0.880 for cognitive features and to 0.790 for MRI images.

From the study of feature importance, we found that the diagnosis is the most significant feature for the three methods. In general, the three methods give importance to features that have been shown to be related to the disease in the state of the art. Apart from the diagnosis, cognitive features tend to occupy the top positions in different feature sets. From the features used in ADNI for establishing the diagnosis, CDRSB is shown to be the most relevant cognitive feature. Focusing on image-based features, the top positions are occupied by measurements from the hippocampus, entorhinal region, and the middle temporal region. Therefore, the methods can identify anatomical structures affected by the disease among more than 700 MRI-based features. Similarly, for PET-based features, FDG-PET measurements in the posterior cingulate gyrus, AV45-PET measurements in the temporal region, or the hypometabolic index are identified as relevant among more than 300 features. These image-based biomarkers have been previously established as relevant in different studies [40–42].

In view of the variability of the XGB-selected features in the top 25 set, we recommend using XGB with model ensembles. Thus, the resulting system would consider the information of the most important features for each ensemble and make better-informed decisions. Meanwhile, our study reported that RF models mainly rely on cognitive information. As it is widely recognized that AD is a multimodal disease, clinically reliable machine learning methods should not be supported only by features from a single domain. The results in this study may indicate that a trustable RF system from the clinical point of view may be modeled with a reduced set of cognitive features to allow other features to increase their importance in the decisions. For SVM, our study reported that the importance given to different features strongly changes among different bootstraps. This result makes us question the suitability of SVM for our computer-aided diagnosis problem. At this point, our study has provided sufficient insights to answer our initial questions:

- Why did the best methods achieve the best accuracy metrics? The best methods consistently used the diagnosis in the feature set. Some of the preprocessing decisions and methodological modifications boosted the mAUC beyond 0.90. Not including the diagnosis in the feature set limits the value of mAUC to 0.89 for the best-performing method.
- How can we quantify the contribution of each feature toward achieving the best accuracies? The built-in importance of tree-based methods and SHAP was shown to be useful for quantifying the contribution of the features toward the accuracies. SHAP provides different ways of representing information that further elucidate feature importance.
- Which features were the most meaningful for the best methods? The diagnosis was the most meaningful feature, with differences among the three methods. Different subsequent features arise depending on the method. For XGB, these features are cognitive features mixed with MRI image-based features. For RF, these features are mostly cognitive. For SVM, cognitive features are highly interleaved with a variety of image-based features.
- Do the best methods use information coherent with clinical knowledge? The violin plot representation of SHAP provided a way to verify that the values of the features are used consistently with clinical knowledge by different systems. Even if the model is built from a large image-based feature set, the methods have been shown to be able to select the most relevant features for making informed decisions.

One shortcoming of our study, inherited from the original TADPOLE Challenge setup, is that the test and evaluation sets originate from the same data distribution used for training the models. Thus, the evaluation and interpretability results of our models may not be preserved with different test and evaluation datasets. Therefore, our findings should be corroborated using different external clinical datasets.

In summary, our study showed the ability of the top two TADPOLE Challenge methods to correctly identify the features most relevant to the prognosis of disease status, thereby boosting our confidence in the potential of these systems for computer-aided prognosis. In the future, we will further explore SHAP features for the improvement of the explanations toward improved interpretability, causability, and actionable explainability. Furthermore, we will extend the study to other TADPOLE Challenge methods with the potential to better adapt to the temporal nature of the prognosis problem that scored lower than the best-performing methods (e.g., *CBIL*) with the aim of establishing a fair ranking within a consistent framework. In addition, we will replace the image-based features with the original images to study the interpretability of the resulting systems in order to build systems capable of extracting useful high-level features invisible to the human eye.

Supporting information

S1 File. Accompanying document with some results complementing the study. Referred in the manuscript as supplementary material. (PDF)

S1 Data. Frequency of the top 25 features over 10 experiments for the 100% training set and the different feature sets. File obtained with XGB. (CSV)

S2 Data. Frequency of the top 25 features over 10 experiments for the 100% training set and the different feature sets. File obtained with RF. (CSV)

S3 Data. Frequency of the top 25 features over 10 experiments for the 100% training set and the different feature sets. File obtained with SVM. (CSV)

Acknowledgments

The authors would like to acknowledge the anonymous reviewers for their valuable revision of the manuscript.

Data used in the preparation of this article were obtained from the Alzheimer's Disease Neuroimaging Initiative (ADNI) database (adni.loni.usc.edu). As such, the investigators within the ADNI contributed to the design and implementation of ADNI and/or provided data but did not participate in the analysis or writing of this report. A complete listing of ADNI investigators can be found at: http://adni.loni.usc.edu/wp-content/uploads/how_to_apply/ADNI_Acknowledgement_List.pdf.

Author Contributions

Conceptualization: Monica Hernandez.

Data curation: Francisco Ferraz.

Formal analysis: Monica Hernandez.

Funding acquisition: Monica Hernandez.

Investigation: Monica Hernandez.

Methodology: Monica Hernandez, Ubaldo Ramon-Julvez, Francisco Ferraz.

Software: Monica Hernandez, Francisco Ferraz.

Supervision: Monica Hernandez.

Writing – original draft: Monica Hernandez.

Writing - review & editing: Ubaldo Ramon-Julvez.

References

- 1. Burns A, Iliffe S. Alzheimer's disease. Brit Med J. 2009;. https://doi.org/10.1136/bmj.b158
- Reeve A, Simcox E, Turnbulla D. Ageing and Parkinson's disease: Why is advancing age the biggest risk factor? Ageing Res Rev. 2014; 14(100):19–30. https://doi.org/10.1016/j.arr.2014.01.004 PMID: 24503004
- Guerreiro R, Bras J. The age factor in Alzheimer's disease. Genome Med. 2015; 7(106). <u>https://doi.org/10.1186/s13073-015-0232-5</u> PMID: 26482651
- Adejare A. Drug Discovery Approaches for the Treatment of Neurodegenerative Disorders. Academic Press, Elsevier. 2017;.
- Piemontese L. New approaches for prevention and treatment of Alzheimer's disease: a fascinating challenge. Neural Regen Res. 2017; 12(3):405–406. https://doi.org/10.4103/1673-5374.202942 PMID: 28469653
- 6. Rasmussen J, Langerman H. Alzheimer's Disease. Why We Need Early Diagnosis. Degener Neurol Neuromuscul Dis. 2019; 9:123–130. https://doi.org/10.2147/DNND.S228939 PMID: 31920420
- Prescott JW. Quantitative Imaging Biomarkers: The Application of Advanced Image Processing and Analysis to Clinical and Preclinical Decision Making. J Digit Imaging. 2013; 26(1):97–108. https://doi. org/10.1007/s10278-012-9465-7 PMID: 22415112

- Beach TG, Monsell SE, Phillips LE, Kukull W. Accuracy of the clinical diagnosis of Alzheimer's disease at National Institute on Aging Alzheimer's Disease Centers. J neuropathology and experimental neurology. 2012; 71(4):266–273. https://doi.org/10.1097/NEN.0b013e31824b211b
- Cure S, Abrams K, Belger M, DellAngnello G, Happich M. Systematic literature review and meta-analysis of diagnostic test accuracy in Alzheimer's disease and other dementia using autopsy as standard of truth. J Alzheimers Dis. 2014;169–82. https://doi.org/10.3233/JAD-131559 PMID: 24840572
- Hua X, Leow AD, Parikshak N, Lee S, Chiang MC, Toga AW, et al. Tensor-based morphometry as a neuroimaging biomarker for Alzheimer's disease: an MRI study of 676 AD, MCI, and normal subjects. Neuroimage. 2008; 43(3):458–469. <u>https://doi.org/10.1016/j.neuroimage.2008.07.013</u> PMID: 18691658
- Ebrahimighahnavieh A, Luo S, Chiong R. Deep learning to detect Alzheimer's disease from neuroimaging: A systematic literature review. Computer Methods and Programs in Biomedicine. 2020; 187:105242. https://doi.org/10.1016/j.cmpb.2019.105242 PMID: 31837630
- Wen J, Thibeau-Sutre E, Diaz-Melo M, Samper-Gonzalez J, Routier A, Bottani S, et al. Convolutional neural networks for classification of Alzheimer's disease: Overview and reproducible evaluation. Med Imag Anal. 2020; 63:101694. https://doi.org/10.1016/j.media.2020.101694 PMID: 32417716
- Bron EE, Smits M, VanDerFlier WM, Vrenken H, Barkhof F, et al. Standardized evaluation of algorithms for computer-aided diagnosis of dementia based on structural MRI: the CADDementia challenge. Neuroimage. 2015; 111:562–579. https://doi.org/10.1016/j.neuroimage.2015.01.048 PMID: 25652394
- Allen GI, Amoroso N, Anghel C, Balagurusamy V, Bare CJ, et al. Crowd-sourced estimation of cognitive decline and resilience in Alzheimer's disease. Alzheimer's & Dementia. 2016; 12(6):645–653. <u>https:// doi.org/10.1016/j.jalz.2016.02.006</u>
- Marinescu RV, Oxtoby NP, Young AL, Bron EE, Toga AW, et al. Tadpole Challenge: prediction of longitudinal evolution in Alzheimers disease. ArXiv. 2018;.
- Marinescu RV, Oxtoby NP, Young AL, Bron EE, Toga AW, et al. The Alzheimer's Disease Prediction Of Longitudinal Evolution (TADPOLE) Challenge: Results after 1 Year Follow-up. ArXiv. 2020;.
- Moore PJ, Lyons TJ, Gallacher J, ADNI. Random forest prediction of Alzheimer's disease using pairwise selection from time series data. PLOS ONE. 2019; p. 1–14. <u>https://doi.org/10.1371/journal.pone.0211558</u> PMID: 30763336
- Adadi A, Berrada M. Peeking Inside the Black-Box: A Survey on Explainable Artificial Intelligence (XAI). IEEE Access. 2017; 6:52138–52160. https://doi.org/10.1109/ACCESS.2018.2870052
- 19. Tjoa E, Guan C. A Survey on Explainable Artificial Intelligence (XAI): Toward Medical XAI. IEEE Transactions on Neural Networks and Learning Systems. 2020;.
- Molnar C. A guide for making black box models explainable. <u>https://christophmgithubio/interpretable-ml-book.</u> 2021;.
- Sundararajan M, Najmi A. The Many Shapley Values for Model Explanation. International Conference on Machine Learning. 2020; p. 9269–9278.
- Merrick L, Taly A. The explanation game: explaining machine learning models using shapley values. Machine Learning and Knowledge Extraction. 2020; p. 17–38.
- Nguyen M, He T, An L, Alexander DC, Feng J, Yeo T, et al. Predicting Alzheimer's disease progression using deep recurrent neural networks. Neuroimage. 2020; 222:117–203. https://doi.org/10.1016/j. neuroimage.2020.117203 PMID: 32763427
- ElSappagh S, Alonso JM, Riazul SM, Sultan AM, Kwak KS. A multilayer multimodal detection and prediction model based on explainable artificial intelligence for Alzheimer's disease. Nature Scientific Reports. 2021; 11:2660. https://doi.org/10.1038/s41598-021-82098-3
- Pfeifer B, Saranti A, Holzinger A. Network module detection from multi-modal node features with a greedy decision forest for actionable explainable AI. ArXiv. 2021;.
- Petersen RC, et al. Alzheimer's Disease Neuroimaging Initiative (ADNI): Clinical characterization. Neurol. 2010; 74(3):201–209. https://doi.org/10.1212/WNL.0b013e3181cb3e25
- Chen T, Guestrin C. XGBoost: A scalable tree boosting system. ACM SIGKDD International Conference on Knowledge Discovery and Data Mining. 2016; p. 785–794.
- Holzinger A, Malle B, Saranti A, Pfeifer B. Towards multi-modal causability with Graph Neural Networks enabling information fusion for explainable AI. Information Fusion. 2021; 71:28–37. <u>https://doi.org/10.1016/j.inffus.2021.01.008</u>
- 29. Pearl J. Causality: Models, Reasoning, and Inference. Cambridge University Press. 2009;.
- Holzinger A, Carrington A, Muller H. Measuring the quality of explanations: The system causability scale (SCS). Comparing human and machine explanations. German J Artif Intell. 2020; p. 193–198. https://doi.org/10.1007/s13218-020-00636-z PMID: 32549653

- Lundberg SM, Lee SI. A unified approach to interpreting model predictions. Proc of the 31st Conference on Neural Information Processing Systems. 2017;.
- Lundberg SM, Erion GG, Lee SI. Consistent Individualized Feature Attribution for Tree Ensembles. ArXiv. 2018;.
- **33.** Ribeiro MT, Singh S, Guestrin C. Why should I trust you?: Explaining the predictions of any classifier. Proc of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining. 2016; p. 1135–1144.
- Herholz K, et al. Discrimination between Alzheimer dementia and controls by automated analysis of multicenter FDG PET. Neuroimage. 2002; 17:302–316. https://doi.org/10.1006/nimg.2002.1208 PMID: 12482085
- Landau SM, et al, ADNI. Associations between cognitive, functional, and FDG-PET measures of decline in AD and MCI. Neurobiol Aging. 2011; 32(7):1207–1218. <u>https://doi.org/10.1016/j.</u> neurobiolaging.2009.07.002 PMID: 19660834
- Chen K, et al. Characterizing Alzheimer's disease using a hypometabolic convergence index. Neuroimage. 2011; 56:52–60. https://doi.org/10.1016/j.neuroimage.2011.01.049 PMID: 21276856
- 37. Kaufman DM, Milstein MJ. Dementia. Clinical neurology for psychiatrists (8th edition). 2017;.
- Ansart M, et al. Predicting the progression of mild cognitive impairment using machine learning: A systematic, quantitative and critical review. Med Image Anal. 2021; 67:101848. https://doi.org/10.1016/j.media.2020.101848 PMID: 33091740
- Spasov SE, Passamonti L, Duggento A, Lio P, Toschi N, ADNI. A parameter-efficient deep learning approach to predict conversion from mild cognitive impairment to Alzheimer's disease. Neuroimage. 2019; 189:276–287. https://doi.org/10.1016/j.neuroimage.2019.01.031 PMID: 30654174
- Dukart J, Kherif F, Mueller K, Adaszewski S, Schroeter ML, Frackowiak RSJ, et al. Generative FDG-PET and MRI model of aging and disease progression in Alzheimer's disease. PLoS Comput Biol. 2013;. https://doi.org/10.1371/journal.pcbi.1002987 PMID: 23592957
- Zhang Y, Dong Z, Phillips P, Wang S, Ji G, Yang J, et al. Detection of subjects and brain regions related to Alzheimer's disease using 3D MRI scans based on eigenbrain and machine learning. Front Comput Neurosci. 2015;. https://doi.org/10.3389/fncom.2015.00066 PMID: 26082713
- Shivamurthy VKN, Tahari AK, Marcus C, Subramaniam RM. Brain FDG PET and the diagnosis of dementia. AJA Am J Roentgenol. 2015; 1:76–85. https://doi.org/10.2214/AJR.13.12363 PMID: 25539279